Author's response to reviews

Title: Baropodometry on Women Suffering From Chronic Pelvic Pain - a cross-sectional study

Authors:

Carolina Kaercher (ckaercher@hotmail.com)
Vanessa K Genro (vanessagenro@gmail.com)
Carlos A Souza (souza.cab@gmail.com)
Mariane Alfonsin (marianealfonsin@ibest.com.br)
Greice Berton (g.berton@terra.com.br)
Joao S Cunha Filho (sabino@vi-rs.net)

Version: 3 Date: 1 November 2011

Author's response to reviews: see over
November 1, 2011.

Prof. Jigisha Patel,
Editor-in-Chief
BMC Womens Health

Dear Prof,

We would like to thank you and the referees for the careful reading and suggestions to our paper entitled “Baropodometry On Women Suffering From Chronic Pelvic Pain – a cross sectional study” MS: 8621039795681698. We performed a new revision of our text and we hope that the final reviewing process finds the manuscript acceptable for publication in the journal.

Answers to Referee 1:

1. Second paragraph was rewritten and elucidated.
2. Sentences were corrected.
3. Tables were corrected and titles were rewritten.

Answers to Referee 2:

Compulsory Revisions

1. Pain length was included in article and we agree that this information is very important to evaluated and compare our results.
2. A new reference was included and discussion topic was amplified to consider influences of movement, respiration on results.
3. Our inclusion criteria and CPP definition was similar to others studies. We included the reference.
4. Sentence on page 4 was rewritten to correctly address our statement and the correct reference was included.
5. Baropodometry influence was discussed in pain evaluation. We stress the importance of a multidisciplinary approach in CPP patients.

Sincerely
Carlos Souza,
Hospital de Clinicas de Porto Alegre
Serviço de Ginecologia e Obstetrícia
Rua Ramiro Barcellos, 2350- 11° andar
Porto Alegre – RGS – Brazil
FAX: +55-51-33300034
Email: souza.cab@gmail.com